Author's response to reviews

**Title:** Differences in Risk Factors for Children With Special Health Care Needs (CSHCN) Receiving Needed Speciality Care by Socioeconomic Status

**Authors:**
- Kristine A Lykens (klykens@hsc.unt.edu)
- Kimberly G Fulda (kfulda@hsc.unt.edu)
- Sejong Bae (sbae@hsc.unt.edu)
- Karan P Singh (ksingh@hsc.unt.edu)

**Version:** 9  **Date:** 13 May 2009

**Author's response to reviews:** see over
May 13, 2009

Joseph Dunckley  
Associate Editor  
BioMedical Central  
*BMC Pediatrics*  

Dear Dr. Dunckley:

Thank you for the opportunity to make revisions our manuscript entitled “Differences in Risk Factors for Children With Special Health Care Needs (CSHCN) Receiving Needed Specialty Care by Socioeconomic Status (MS:3173576532175765).

We have carefully reviewed the abstract and manuscript formatting requirements and have made the following changes to our submission; 1) reformatted the abstract to conform to journal requirements, 2) revised the title page to conform to journal requirements, 3) revised the statement of no competing interests, 4) added the statement that all authors have read the final manuscript, and 5) made punctuation revisions to the reference section to conform with the example given.

Sincerely,

Kristine Lykens